

AGU Advances

Peer Review History of

The habitat of the nascent Chicxulub crater

T. J. Bralower¹, J. Cosmidis¹, M. S. Fantle¹, C. M. Lowery², B. H. Passey³, S. P. S. Gulick², J. V. Morgan⁴, V. Vajda⁵, M. T. Whalen⁶, A. Wittmann⁷, N. Artemieva⁸, K. Farley⁹, S. Goderis¹⁰, E. Hajek¹, P. J. Heaney¹, D. A. Kring¹¹, S. L. Lyons¹, C. Rasmussen², E. Sibert¹², F. J. Rodríguez Tovar¹³, G. Turner-Walker¹⁴, J. C. Zachos¹⁵, J. Carte¹, S. A. Chen¹, C. Cockell¹⁶, M. Coolen¹⁷, K. H. Freeman¹, J. Garber¹, M. Gonzalez¹, J. L. Gray¹⁸, K. Grice¹⁷, H. L. Jones¹, B. Schaefer¹⁷, J. Smit¹⁹, S. M. Tikoo²⁰

¹Department of Geosciences, Pennsylvania State University, University Park, PA, 16802, USA ²Institute for Geophysics, Jackson School of Geosciences, University of Texas at Austin, Austin, TX 78758, USA

³Department of Earth and Environmental Sciences, University of Michigan, Ann Arbor, MI 48109-1005, USA

Department of Earth Science and Engineering, Imperial College London, UK
 Department of Palaeobiology, Swedish Museum of Natural History, Stockholm, Sweden
 Geophysical Institute, University of Alaska, Fairbanks, AK 99775, USA
 Eyring Materials Center, Arizona State University Tempe, AZ 85287-8301, USA
 Planetary Science Institute, Tucson, AZ, USA

⁹Division of Geological and Planetary Sciences, California Institute of Technology, Pasadena, CA 91125, USA

¹⁰Department of Chemistry, Vrije Universiteit Brussel, BE-1050 Brussels, Belgium
 ¹¹Lunar and Planetary Institute, 3600 Bay Area Blvd., Houston, TX 77058-1113, USA
 ¹²Department of Earth and Planetary Sciences, Yale University, New Haven, CT, 06520, USA
 ¹³Departamento de Estratigrafía y Paleontología, Facultad de Ciencias, Universidad de Granada, Granada 18002, Spain

¹⁴Graduate School of Cultural Heritage Conservation, National Yunlin University of Science and Technology, 640 Yunlin, Taiwan

¹⁵Earth and Planetary Sciences, University of California Santa Cruz, 1156 High Street, Santa Cruz, CA 95064, USA

¹⁶School of Physics and Astronomy, University of Edinburgh, Edinburgh, EH9 3FD, UK
 ¹⁷Organic and Isotope Geochemistry Centre, The Institute for Geoscience Research, School of Earth and Planetary Science, Curtin University, Perth, WA, Australia

¹⁸Materials Research Institute, Pennsylvania State University, University Park, PA, 16802, USA
¹⁹Department of Geology and Geochemistry, VU University Amsterdam 1081HV Amsterdam, The Netherlands

²⁰Department of Geophysics, Stanford University, Stanford, CA 94305, USA

Files Uploaded Separately

Original Version of Manuscript (2020AV000208)
First Revision of Manuscript [Accepted] (2020AV000208R)

Peer Review Comments on 2020AV000208

Reviewer #1

General Comments:

The Bralower et al revised manuscript on a new Chicxulub crater drill core presents a variety of new information to reconstruct the paleoenvironment of the Chicxulub crater in the subsequent months to years to tens of thousands of years after the crater was formed. Overall I believe this to be a much improved version of the original submission that the multiple lines of evidence about the habitat of this impact crater in the aftermath of its formation is much more connected together. Specifically I appreciated the expansions of sections 3.3, 3.4, and 3.5 which go a long way in connecting more of the huge variety of details in this manuscript. Additionally, the new figures made and new versions of the original figures are clearer and more understandable, in the context of internal consistency with the manuscript text but more easy to follow from one figure to the next. I only have a couple minor comments worth considering.

Specific Comments

(line numbers refer to clean version of resubmission)

Abstract (Line 73): I suppose you are referring to the tsunami deposits themselves here in the 'hours' end of the spectrum, as the rest of the manuscript and figures refers to the seiche deposits being formed within days to years.

Introduction (Line 119): how can the authors claim that LIP volcanism played no part, 'unrelated' in the K-Pg mass extinction? Many other LIP eruptions in the Phanerozoic have been linked to mass extinctions, most famously is the Siberian Traps and the EPME...why does Deccan have no relationship? These are geologically unique phenomenon just as large bolide impacts, and I just don't see how any geoscientist can simply state that either of them would have no impact on the existing biosphere at the time as there is a whole host of biogeochemical-Earth System feedbacks well established with these types of explosive volcanism.

-Seems more appropriate, less conjectural, to say that the volcanism is unrelated to the Chicxulub impact event. There are a whole host of papers that have drawn links to the Deccan Traps and this mass extinction... seems unnecessarily dismissive of this hypothesis even though it has no bearing on your data and story presented here.

Reviewer #2

Reviewed by R. Mark Leckie, Univ. of Massachusetts Amherst

This is a well-written, thoroughly documented analysis of the biosphere response to the K-Pg impact as recorded in a remarkable succession of sedimentary deposits from the peak ring of the crater. The new stable isotope and clumped isotope paleotemperature data nicely track the biotic succession recovered in the core; the interpretation that the cooler, elevated peak ring was inhabited by survivor nannoplankton and a monospecific dinocyst assemblage illustrates how much fine detail is preserved in this record. Calcareous dinocysts in association with charcoal and the Ir-anomaly, a widely observed association, provides strong support for the hypothesis of darkness (and cooling) in the immediate aftermath of the impact. The likely implication of very rapid evolution of the earliest Paleocene planktic forams, consistent with the 3He data, is very compelling and significant.

The paper is ready for publication, but I have a couple of minor questions, a suggestion, and a typo.

Line 62: First key point: How can tsunami and seiche waves last months to years?: Hours to days seems more likely; how/why would such waves last any longer? Can you provide support/documentation for such protracted activity following a singular abrupt event?

Line 81: Were the seiche triggered by the impact-generated tsunami, or by winds? I think it would be helpful for the reader to understand how you are differentiating between the tsunami deposit and seiche deposits. The duration of a tsunami would have been hours to days, right? Is an extreme perturbed climate, like the modeled super cooling that may have lasted for decades (Brugger et al. 2016) responsible for winds to drive a seiche during the years post-tsunami?

Line 110: 'In one of the most rapid geomorphic events in Earth history...' Perhaps of of the most rapid geomorphic, biotic and environmental events in Earth history.

Line 446: The Ir anomaly generally signifies fallout within years of the impact (Not with years of the impact).

Reviewer #3

This paper presents an impressive amount of new data on the Chicxulub crater IODP core. It is well written and the interpretations are well supported by the data. The sedimentological and clumped isotope data are novel and provide important constraints on the conditions after the impact. The authors have carefully taken in consideration the reviewer comments and modified the manuscript accordingly.

The manuscript provides a compelling story which is well supported by the data, even if in parts its speculative. It will definitely stimulate the discussion on the aftermath of the Cicxulub inpact in the literature and can be published basically as is.

I only have a few minor comments

Line 551: should be "per meg" not "per mg"

Supplementary. Information:

A table with the carbon, oxygen and clumped isotopes should be added to the supplement.

The authors should cite Bernasconi et al. 2018 as source information for the ETH standards, for which they report different values that the "accepted" values used by other laboratories. These standards have been used in producing different temperature calibrations from those used in this paper and their values may change. It is important to report the original publication with the current accepted values if other workers wish to recalculate the temperatures with a different D47-Temperature equations and recalculate the data according to different equations in the future.

Bernasconi S.M., Müller I. A., Bergmann K.D., Breitenbach S. F. M., Fernandez, A., Hodell D. A., Jaggi M., Meckler A.N. Millan I., and Ziegler M. (2018). Reducing uncertainties in carbonate

clumped isotope analysis through consistent carbonate-based standardization. Geochemistry,

Geophysics, Geosystems, 1-20.